Author’s response to reviews

Title: Attempting rigour and replicability in thematic analysis of qualitative research data; a case study of codebook development

Authors:
Kate Roberts (kate.roberts@postgrad.otago.ac.nz)
Anthony Dowell (tony.dowell@otago.ac.nz)
Jing-Bao Nie (jing-bao.nie@otago.ac.nz)

Version: 1 Date: 17 Jan 2019

Author’s response to reviews:

BMRM-D-18-00402
Demonstrating rigour in thematic analysis; a case study of code book development
Kate Roberts; Tony Dowell; Jing-Bao Nie
BMC Medical Research Methodology

Dear DrLivia Puljak,

Thank you for the opportunity to resubmit the afore mentioned article. The comments from both yourself and the reviewers were extremely valuable and indeed have enabled me to strengthen this piece of work.

Please find attached the two copies of the revised article which has been updated to reflect the advice from all reviewers. One version shows all the changes made using track changes and the other is a clean final version. In addition the I have submitted a separate document outlining summary of changes made.

Best wishes,

Kate Roberts